Author's response to reviews

Title: Serotonin transporter gene polymorphism (5-HTTLPR) L allele interacts with stress to increase anxiety symptoms in Chinese adolescents: a multiwave longitudinal study

Authors:

Qingsen Ming (mingqingsen@163.com)
Yun Zhang (zhangyun917@163.com)
Jinyao Yi (jinyaoyi@gmail.com)
Xiang Wang (wang0916xia@gmail.com)
Xiongzhao Zhu (xiongzhaozhu@163.com)
Shuqiao Yao (shuqiaoyao@163.com)

Version: 2 Date: 27 May 2015

Author's response to reviews: see over
Dear Prof. Chua,

Thank you very much for your suggestions for revision. We have revised our manuscript as following:

1. We included line numbers in each page.
2. We included page numbers.
3. We removed all the page breaks.

Thank you for your time and processing our manuscript.

Sincerely,

Shuqiao Yao, M.D., Ph.D.
Professor of Psychiatry and Psychology
Medical Psychological institute
The Second Xiangya Hospital of Central South University
Changsha, Hunan 410011,
P.R. China
Phone: 86-731-85292130
FAX: 86-731-85361328
E-mail: shuqiaoyao@163.com